%% LaTeX version of the Royal Society grant application form
%% form desgin belongs to the Royal Soceity
%%
%% This LaTeX code is GPL- you can distribute freely but cannot
%% charge for it, you can change it but this GPL remains intact
%%
%% (c) Matt Clark June 2002 
%% (matt dot clark at nottingham dot ac dot uk - no spam)
%%
%% most of what needs to be filled in has PLACEHOLDERS in capitals
%% this form has been previously accepted by the RS without comment


\documentclass[a4paper,8pt]{article}
\usepackage{epsfig}
%% the following sizes are the biggest sensible 
%\textwidth 17.3cm
%\textheight 25.7cm
\pagestyle{empty}
\topmargin -1.5cm
\oddsidemargin=-1.2cm
\leftmargin 0cm
\textwidth 19cm
\textheight 267cm
\begin{document}
\renewcommand\refname{\vspace{-1cm}}
\bibstyle{unsrt}
\bibliographystyle{unsrt}  
\begin{center}
{\large \bf Application for a Royal Society Research Grant}
\end{center}

\noindent To be typewritten\hspace{\fill} Ref:RSRG(E)\\

\noindent\begin{tabular}{p{3in}p{2in}}
1\hspace{0.3cm} Title: \hspace{1cm} Forenames (in full):  &Surname:  \\
\hspace{1.5cm}  TITLE \hspace{3.5cm}  FORENAME &\hspace{2cm} SURNAME  \\\\
{ \raggedright
2\hspace{0.3cm}  Address for correspondence:\\
\hspace{0.5cm} ADDRESS 1\\
\hspace{0.5cm} ADDRESS 2\\
\hspace{0.5cm} ADDRESS 3\\
\hspace{0.5cm} ADDRESS 4\\
\hspace{0.5cm} ADDRESS 5\\
\hspace{0.5cm} ADDRESS 6\\}
&
{\raggedright 
Tel: NUMBER\\
(daytime)\\
extn: EXTENSION\\
\vspace{0.5cm}
email:\\
{\tt EMAIL ADDRESS}\\}\\
\end{tabular}

\noindent3\indent Current post: JOB TITLE
\\

%% use this format to circle option -> \textcircled{A}
Subject:\hspace{1cm}A B C D E F G H I 
\\

University/other establishment: LOCATION\\

\noindent4\indent Brief title of Project (MUST NOT EXCEED 34 CHARACTERS AND SPACES IN LENGTH)\\

%% Ok now the next bit is silly- but it does work - matt
%% the \em in the empty boxes keep the box spacing sane
{\tt\small 
\begin{tabular}{|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|p{.02in}|}
\hline
Y&O&U&R&\em&T&I&T&L&E&\em&H&E&R&E&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\em&\\
\hline
\end{tabular}
}\\

\noindent5\indent Sum requested (including VAT if applicable) \pounds POUNDS (to nearest \pounds)\\

Date needed: Day DD Month MM Year YYYY\\

\noindent6\indent Breakdown of sum requested, including VAT (total should equal sum given in 5):\\

\begin{tabular}{p{4in}r}
ITEM HERE & \pounds POUNDS \\ 
ITEM HERE & \pounds POUNDS \\ 
ITEM HERE & \pounds POUNDS \\ 
ITEM HERE & \pounds POUNDS \\ 
ITEM HERE & \pounds POUNDS \\ 
\end{tabular}     \\
\\
\\
\\

\noindent7\indent Other support (in \pounds sterling) $[$ for the project for which the Royal Society is sought$]$\\

\begin{tabular}{p{1.2in}p{1.2in}lp{3in}}
a) Assured  &\pounds POUNDS &Source(s)& SOURCE\\\\

b) Applied for& \pounds POUNDS &Source(s)& SOURCE\\
\end{tabular}\\

\noindent8\indent Approval of the Institution at which the research will be conducted (to support application and accept any award \\\indent under the normal terms and conditions relating to Royal Society Grants):\\

a) Signature of the Head of Department:\\

b) Signature/stamp of Institution:\\
\\

{\small Where any of the proposed research involves the use of animals I have read and confirm I am in compliance 
with the Royal Society's\\\indent policy on the funding of research involving animals and the publication of the results of such research. (\underline{Available on request or at} \\
\indent\underline{http://www.royalsoc.ac.uk/st\_pol48.htm})}\\

\noindent9\indent Date: 20 January 2001 \hspace{5cm} Signature of applicant:\\

\noindent For Office Use Only\\
\hspace*{2.5cm} A Application received (dd/mm/yy) \hspace{5.5cm} RSRG\\

\noindent B Board assigned \hspace{2cm} C Grant Recommended \hspace{1cm}\pounds\hspace{4.5cm}Grade\\

\noindent D Grant awarded \hspace{2cm}\pounds	\hspace{4cm} Date CM (dd/mm/yy)\\
\newpage
\noindent10\indent Proposed project: (400 word maximum)\\

\noindent\begin{tabular}{|p{7in}|}\hline
YOUR PROPOSED PROJECT HERE
%% Put you proposal text here
%%
%% adjust the argument in the vspace below to vary the whitespace at
%% the end of this box
%% if require
\vspace*{6cm}\\
\hline
\end{tabular}\\

\noindent11\indent Brief statement of scientific career including qualifications and dates obtained:
\\
%% put your stuff here - it should take up 40-50mm at the most to fit in with the original
\\
\\
\\
\\
\noindent12\indent Main recent publications: (to a maximum of 5)
%% I personally use bibtex to generate this with a command
%% like
%% \cite{my,five,publications,relevant,here}
%% {\small \bibliography{my_bibliography_file}}
%% I then latex the document and bibtex it, this build the .bbl file which is included here
%% this gives the citation marks [1,2,3,4,5] at this point so I then comment out the
%% \cite command and re-latex as long as I don't rebibtex the references remain with
%% no citation marks--- matt

\noindent{\small Personal data on this form will be held and processed on the Society's computer.  A summary of the Society's data protection policy, including the rights of subjects upon whom data is held, is obtainable from the Executive Secretary (ref: DPSA/JHS).}


\end{document}



